

# 

**Citation:** Giliberti R, Cavaliere S, Mauriello IE, Ercolini D, Pasolli E (2022) Host phenotype classification from human microbiome data is mainly driven by the presence of microbial taxa. PLoS Comput Biol 18(4): e1010066. https://doi. org/10.1371/journal.pcbi.1010066

**Editor:** Luis Pedro Coelho, Fudan University, CHINA

Received: October 13, 2021

Accepted: March 29, 2022

Published: April 21, 2022

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pcbi.1010066

**Copyright:** © 2022 Giliberti et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data and source code used to produce the results and analyses presented in this manuscript are available on a

RESEARCH ARTICLE

# Host phenotype classification from human microbiome data is mainly driven by the presence of microbial taxa

Renato Giliberti<sup>1</sup>, Sara Cavaliere<sup>1</sup>, Italia Elisa Mauriello<sup>1</sup>, Danilo Ercolini<sup>1,2</sup>, Edoardo Pasolli<sup>0</sup><sup>1,2</sup>\*

1 Department of Agricultural Sciences, University of Naples Federico II, Portici, Italy, 2 Task Force on Microbiome Studies, University of Naples Federico II, Naples, Italy

\* edoardo.pasolli@unina.it

# Abstract

Machine learning-based classification approaches are widely used to predict host phenotypes from microbiome data. Classifiers are typically employed by considering operational taxonomic units or relative abundance profiles as input features. Such types of data are intrinsically sparse, which opens the opportunity to make predictions from the presence/ absence rather than the relative abundance of microbial taxa. This also poses the question whether it is the presence rather than the abundance of particular taxa to be relevant for discrimination purposes, an aspect that has been so far overlooked in the literature. In this paper, we aim at filling this gap by performing a meta-analysis on 4,128 publicly available metagenomes associated with multiple case-control studies. At species-level taxonomic resolution, we show that it is the presence rather than the relative abundance of specific microbial taxa to be important when building classification models. Such findings are robust to the choice of the classifier and confirmed by statistical tests applied to identifying differentially abundant/present taxa. Results are further confirmed at coarser taxonomic resolutions and validated on 4,026 additional 16S rRNA samples coming from 30 public case-control studies.

# Author summary

The composition of the human microbiome has been linked to a large number of different diseases. In this context, classification methodologies based on machine learning approaches have represented a promising tool for diagnostic purposes from metage-nomics data. The link between microbial population composition and host phenotypes has been usually performed by considering taxonomic profiles represented by relative abundances of microbial species. In this study, we show that it is more the presence rather than the relative abundance of microbial taxa to be relevant to maximize classification accuracy. This is accomplished by conducting a meta-analysis on more than 4,000 shot-gun metagenomes coming from 25 case-control studies and in which original relative abundance data are degraded to presence/absence profiles. Findings are also extended to

#### GitHub repository at https://github.com/RGilib/ giliberti-meta-analysis-2022.

**Funding:** The work was by supported by P.O.R. Campania FSE 2014/2020 to R.G. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

16S rRNA data and advance the research field in building prediction models directly from human microbiome data.

This is a PLOS Computational Biology Methods paper.

# Introduction

Evidence has linked the human microbiome, the large set of microorganisms that reside in our body, with health and disease conditions [1]. Several diseases have been associated with microbiome traits and estimation of host phenotypes from microbiome composition has received remarkable attention in the community. In this regard, growing attention has been given to predicting host phenotypes using machine-learning based approaches, and in which adoption of classification methodologies for case-control studies has represented the most investigated scenario [2]. Classification represents a practical approach to implicitly integrate multiple characteristics (i.e., features; such as the case of combination of hundreds of microbial relative abundances) and get evaluation metrics of relatively easy interpretation. This is the case of the area under the receiver operating characteristic curve (AUC), the most used metric in the microbiome field for binary classification problems [2], which ranges in value from 0 to 1 with better accuracy when moving towards one.

Focusing on case-control studies, machine learning methods have been involved in two main types of analyses. The first has relied on applying established methodologies to newly generated data, which has allowed researchers to provide evidence of the predictability of host phenotypes from microbiome data for several different diseases including inflammatory bowel disease [3], obesity [4], type-2 diabetes [5], colorectal cancer [6], and paved the way to the potential use of the microbiome as a diagnostic tool [7,8]. The increasing number of large population studies [9,10] has also enabled the implementation of several (large-scale) meta-analyses aiming at validating findings across independent cohorts. Besides analyses based on 16S rRNA data [11–13], similar efforts have been extended more recently to shotgun data [14–17], while extension to other-omics data has been more challenging [18]. The second group of analyses has been focused on the proposal of new methodologies in two main directions: extraction of better feature representations or optimization at classifier level [19]. While classification can be applied on the original set of features, improvements can be obtained by reducing the dimensionality of the feature space (for example by selecting or extracting specific operational taxonomic units (OTUs) or microbial taxa). Examples include feature subset selection [20], recursive feature elimination [14], and hierarchical feature engineering [21]. Different (supervised) methods have been adopted for classification purposes. Some widely used strategies are represented by logistic regression [22], support vector machines (SVMs) [3], knearest neighbours [23], and random forests (RFs) [14]. Comparisons among different classifiers have also been performed, with ensemble methods such as RFs and extreme gradient boosting decision trees that have exhibited in general the best performances [24]. Recently, different solutions based on deep learning approaches have been also proposed [25,26], including methods to transform high-dimensional data into robust low-dimensional representations [27], although challenges still arise due to the limited amount of labelled information that is typically available in case-control microbiome studies [28].

Despite the different methodologies adopted along the classification pipeline, classification models have been typically built by considering OTU or relative abundance profiles as input features. However, such types of data are intrinsically sparse, therefore this potentially enables to make inferences from the presence/absence of microbial taxa rather than their relative abundance values. This also poses the question whether it is the presence of particular taxa rather their abundance values to be relevant for discrimination purposes. Surprisingly, this aspect has not been investigated yet. In this paper, we aim at filling this gap by presenting a meta-analysis on publicly available datasets from both shotgun and 16S rRNA data.

# Materials and methods

### The considered publicly available metagenomic and 16S rRNA datasets

In this paper, we conducted a meta-analysis on publicly available human metagenomic datasets for host phenotype classification. More specifically, we considered 4,128 samples coming from 25 shotgun metagenomic studies/datasets as summarized in Table 1 and Fig 1A. Twenty-one studies were devoted to the characterization of the gut microbiome in association with different diseases (i.e., case-control studies). Two additional datasets were case-control studies (peri-implantitis, mucositis, and schizophrenia) from oral metagenomes. We also considered a dataset aiming at characterizing changes in the human microbiome due to

Table 1. Summary of the 25 classification tasks derived from metagenomic datasets for case-control prediction. ACDV: Atherosclerotic cardiovascular disease, AD: Alzheimer's disease, BD: Behcet's disease, CRC: Colorectal cancer, IBD: irritable bowel disease, T1D: Type 1 diabetes, T2D: Type 2 diabetes. We additionally considered the HMP\_2012 dataset [10] for body site discrimination between gut (N = 414) and oral (N = 147) samples.

Dataset name	Body site	# controls	Cases	# cases	Reference
JieZ_2017	Gut	171	ACVD	214	[31]
ChngKR_2016	Skin	40	AD	38	[32]
YeZ_2018	Gut	45	BD	20	[33]
RaymondF_2016	Gut	36	Cephalosporins	36	[34]
QinN_2014	Gut	114	Cirrhosis	123	[35]
FengQ_2015	Gut	61	CRC	46	[36]
GuptaA_2019	Gut	30	CRC	28	[37]
HanniganGD_2017	Gut	28	CRC	27	[38]
ThomasAM_2018a	Gut	24	CRC	29	[39]
ThomasAM_2018b	Gut	28	CRC	32	[39]
VogtmannE_2016	Gut	52	CRC	52	[40]
WirbelJ_2018	Gut	65	CRC	60	[41]
YachidaS_2019	Gut	251	CRC	258	[42]
YuJ_2015	Gut	53	CRC	75	[43]
ZellerG_2014	Gut	54	CRC	61	[6]
LiJ_2017	Gut	41	Hypertension	99	[44]
IjazUZ_2017	Gut	38	IBD	56	[45]
NielsenHB_2014	Gut	248	IBD	148	[46]
GhensiP_2019_m	Oral	49	Mucositis	20	[47]
GhensiP_2019	Oral	49	Peri-implantitis	23	[47]
Castro_NallarE_2015	Oral	16	Schizophrenia	16	[48]
Heitz-BuschartA_2016	Gut	26	T1D	27	[49]
KosticAD_2015	Gut	89	T1D	31	[50]
KarlssonFH_2013	Gut	43	T2D	53	[51]
QinJ_2012	Gut	174	T2D	170	[52]

https://doi.org/10.1371/journal.pcbi.1010066.t001



Fig 1. Classification accuracies are robust to degradation from species-level relative abundance to presence/absence profiles in shotgun datasets. Results obtained on 25 case-control studies for host phenotype classification from human microbiomes. (A) Number of case and control samples across the different studies. (B) AUC and (C) AUPRC scores using RF as back-end classifiers on species-level taxonomic profiles. Comparison between relative abundance (in blue) and presence/absence (in red) profiles highlighted negligible differences and no statistical differences in none of the studies (see S1 Fig for AUC scores and S2 Table for p-values). Metrics of comparison in terms of AUC, AUPRC, precision, recall, and F1 are summarized in S2 Fig and S2 Table is represented a comparison between AUC and AUPCR scores. (D) Number of statistically significant taxa from relative abundance (in blue) and presence/absence (in red) profiles.

https://doi.org/10.1371/journal.pcbi.1010066.g001

consumption of cephalosporins, while the last dataset was devoted to the discrimination between body sites (i.e., stool vs oral) in the Human Microbiome Project (HMP) dataset. Metagenomic samples were processed to generate species-level taxonomic profiles through MetaPhlAn3 [29]. Species abundances are expressed as real numbers in the range [0,1] with values that sum to 1 for each sample. Generation of relative abundances at other taxonomic levels (i.e., genus, family, and order) was also extracted from the MetaPhlAn3 output. Metadata information in terms of disease status or body site for the HMP dataset are available in the curatedMetagenomicData package [30].

We additionally analysed 4,026 16S rRNA samples coming from 30 publicly available casecontrol studies (S1 Table and Fig 2A). We considered the same set of gut samples considered



**Fig 2. Classification accuracies are robust to degradation from genus-level relative abundance to presence/absence profiles in 16S rRNA datasets.** Results obtained on 30 case-control studies for host phenotype classification from human microbiomes. (A) Number of case and control samples across the different studies. (B) AUC and (C) AUPRC scores using RF as back-end classifiers on species-level taxonomic profiles. Comparison between relative abundance (in blue) and presence/absence (in red) profiles highlighted negligible differences and no statistical differences in none of the studies (see <u>S5 Table</u> for p-values) as found also in shotgun datasets (see <u>Fig 1</u>). Metrics of comparison in terms of AUC, AUPRC, precision, recall, and F1 are summarized in <u>S5 Table</u>. (C) Number of statistically significant taxa from relative abundance (in blue) and presence/absence (in red) profiles.

https://doi.org/10.1371/journal.pcbi.1010066.g002

in [13] with metadata information in terms of disease status as follows: autism spectrum disorder (ASD), *Clostridioides difficile* infection (CDI), CRC, enteric diarrheal disease (EDD), human immunodeficiency virus (HIV), IBD, liver cirrhosis (CIRR), minimal hepatic encephalopathy (MHE), non-alcoholic steatohepatitis (NASH), obesity (OB), Parkinson disease, psoriatic arthritis (PSA), rheumatoid arthritis (RA), and T1D. 16S rRNA samples were preprocessed following the same procedure adopted in [13]. More specifically, we discarded samples with fewer than 100 reads and removed OTUs with less than 10 reads and/or present in less than 1% of the samples. After calculating the relative abundance of each OTU, OTUs were collapsed to genus level by summing their relative abundance values and by discarding any OTUs which were un-annotated at the genus level.

### The adopted machine learning methods

The classification tasks on both shotgun and 16S rRNA data were carried out by considering the already developed and validated MetAML (*Met*agenomic prediction Analysis based on *M*achine *L*earning) tool [14]. Main analyses were conducted by using Random Forests (RFs) as back-end classifiers, and validations were extended to other three classifier types: support vector machines with linear (denoted with LSVM in this paper) and RBF (denoted with SVM in this paper) kernel, Lasso, and Elastic Net (ENet).

Free parameters of the classifiers were set as follows. For RF, i) the number of trees was set to 500, ii) the number of features to consider when looking for the best split was equal to the root of the number of original features, and iii) the gini impurity criterion was used to measure the quality of a split. For Lasso and ENet, the regularization parameters were obtained using a 5-fold stratified cross-validation approach. For Lasso the alpha parameter was found in the set  $\{10^{-4}, \ldots, 10^{-0.5}\}$  with 50 uniform steps. For ENet, besides the alpha parameter, also the L1\_ratio parameter was chosen in the set [0.1, 0.5, 0.7, 0.9, 0.95, 0.99, 1.0].

#### Validation and evaluation strategies

We conducted two main types of analysis: i) cross-validation and ii) cross-study analysis. In cross-validation, samples were randomly divided into k (with k = 10 in our case) folds by considering a stratified cross-validation approach to preserve the percentage of samples of each class. Results were repeated and averaged on 20 independent runs. Different models were trained on the same cross-validation splits. We also considered a cross-study analysis in order to evaluate robustness of the prediction when transferring models from a source to a target domain. In this setting, the classification model was trained on the source dataset and accuracy was evaluated on a different independent dataset.

Classification accuracies were evaluated in terms of five main metrics: area under the curve (AUC), area under the precision-recall curve (AUPRC), precision, recall, and F1.

We calculated mean difference and standard error for each 10-fold CV and averaged across the 20 repetitions. We calculated the 95% confidence interval on the difference in AUC performance between two classifiers as done in [14] using the t-distribution with df = 9:

95%CI: 
$$\frac{1}{20} \frac{1}{10} \sum_{j=1}^{20} \sum_{i=1}^{10} (AUC_{1ij} - AUC_{2ij}) \pm 2.26 x \frac{\sigma_j}{\sqrt{10}}$$
 (1)

where AUC1ij and AUC2ij are the AUC of two classifiers in fold i of repetition j, and  $\sigma$ j is the standard deviation of the AUC1ij–AUC2ij across i = 1...10 folds in repetition j. We computed the p-values from the t-statistics from mean difference and standard error flatten over the 20

repetitions:

$$t = \frac{\frac{1}{20} \frac{1}{10} \sum_{j=1}^{20} \sum_{i=1}^{10} (AUC_{1ij} - AUC_{2ij})}{\frac{1}{20} \sum_{j=1}^{20} \frac{\sigma_j}{\sqrt{10}}}$$

We used a two-tailed t-test with df = 9.

#### Experimental setting for shotgun datasets

Most of the analyses on shotgun datasets were conducted by considering a cross-validation approach. Twenty-four classification tasks were devoted to the discrimination of healthy from diseased subjects (i.e., case-control studies), while the HMP dataset was used to perform body site discrimination between gut and oral samples. We also considered the ten independent datasets associated with CRC and evaluated prediction capabilities in a cross-study setting.

Baseline results were obtained by considering the original relative abundance profiles at species-level resolution provided by MetaPhlAn3 [29] as features and using RF as back-end classifier. This is the setting that was successfully deployed and validated in multiple metaanalyses such as the ones presented in [14,30,39,47]. At this point, multiple comparisons were performed: i) starting from the original species-level relative abundance profiles (one profile for each sample), we generated presence/absence profiles by simply thresholding the relative abundance values at 0%. This generated a set of boolean profiles where 1 indicated the presence of the species regardless of its relative abundance in the considered sample, while 0 was associated with its absence. The same approach based on RF was applied on this set of newly generated profiles and compared with the results obtained on the original relative abundances. Results are summarized in Figs 1B, 1C, S1 and S2; ii) the same procedure described in i) was applied again by thresholding the relative abundance profiles at different values to assess sensitivity of classification to low abundant species. We considered these values as threshold levels: 0.0001%, 0.001%, 0.01%, and 0.1%. Results using RF as classifier are summarized in Figs 3A and \$3A; iii) we extended the comparison done at species-level between original relative abundance and boolean (with threshold = 0%) profiles to three other taxonomic levels (i.e., genus, family, and order) to evaluate sensitivity of classification when moving from species to coarser taxonomic resolutions. Results with RF classification are summarized in Figs 4 and S3B; iv) we finally assessed robustness of our findings to the choice of the classification method. We compared RF results with the ones obtained by other four classifier algorithms (i.e., SVM with linear kernel, SVM with RBF kernel, Lasso, ENet) for both relative abundance and presence/absence profiles (Figs 5 and S3C). While we report in main figures only comparisons in terms of AUC, comparisons for the other three metrics (i.e., precision, recall, and F1) are reported in S2 Table.

#### Experimental settings for 16S rRNA datasets

For 16S rRNA datasets we carried out only cross-validation analyses. From the genus-level profiles generated as described in the section "The considered publicly available metagenomic and 16S rRNA datasets", we generated the boolean profiles (with threshold = 0%) as similarly done for shotgun data. We compared the two types of profiles using a RF classifier (results in **Figs 2B**, **2C** and **S4**), and were then extended also to the other classifier types (results in **S3 Table**).

#### Statistical tests

On the same set of scenarios in which we compared classification accuracies, we conducted statistical tests to evaluate to which extent degradation from relative abundance to boolean profiles can impact the identification of differentially abundant species. We used Mann-Whitney U test

(2)



**Fig 3. Classification accuracies are not impacted when relative abundances are thresholded up to 0.001%**. Results on the 25 case-control shotgun studies by comparing the baseline (i.e., species-level relative abundance profiles) with the presence/absence profiles generated by thresholding at different relative abundance values (ranging from 0% to 0.1%). (A) Difference in AUC between the presence/absence and the relative abundance RF classification result. A positive value indicates that presence/absence outperforms relative abundance data. AUC scores at different thresholds are summarized in \$2 Table. (B) Difference in number of statistically significant taxa (numbers summarized in \$7 Table).

https://doi.org/10.1371/journal.pcbi.1010066.g003

to identify the set of significant taxa when relative abundance profiles were involved, while we adopted Fisher exact test to deal with presence/absence data. Although it is out of the scope of the present study to perform a comprehensive evaluation of available statistical tests, further investigation taking into account alternatives including methodologies that can deal with compositional issues [53,54] is warranted. Finally, false detection rate (FDR) was applied for multiple testing correction, and corrected p-values < 0.05 identified significant taxa.



**Fig 4. Classification results are more impacted to relative abundance degradation at coarser taxonomic resolution.** Results on the 25 casecontrol shotgun studies by comparing the baseline (i.e., relative abundance profiles) with the presence/absence profile generated by thresholding at 0.0% and varying taxonomic resolution from species to order level. Difference in AUC between the presence/absence and the relative abundance RF classification result. A positive value indicates that presence/absence outperforms relative abundance data.

https://doi.org/10.1371/journal.pcbi.1010066.g004

#### **Rarefaction analysis**

We further performed rarefaction analysis by: i) considering the three datasets having the highest number of significant species from relative abundance profiles (i.e., JieZ\_2017, Niel-senHB\_2014, and QinN\_2014); ii) rarefying raw reads (using https://github.com/lh3/seqtk) and considering 1M reads for each metagenome; iii) applying the same pipeline to generate taxonomic profiles through MetaPhlAn3; iv) applying the same pipeline to build classification models and identifying statistically significant species.





https://doi.org/10.1371/journal.pcbi.1010066.g005

## **Results and discussion**

In this paper, we conducted a meta-analysis aiming at evaluating to which extent degradation from relative abundance to presence/absence of microbial taxa can impact host phenotype classification from human metagenomes. The analysis was conducted on 4,128 public available metagenomes coming from 25 datasets (Table 1 and Fig 1A). Metagenomes were uniformly processed to generate species-level taxonomic profiles with MetaPhlAn3 [29] (see Materials and Methods) with metadata information available in the curatedMetagenomicData package

[30]. From relative abundance profiles, expressed as real numbers in the range [0, 1], we generated presence/absence profiles by simply thresholding the relative abundance values at 0%. This generated a set of boolean profiles where one indicated the presence of the species regardless of its relative abundance in the considered sample, while zero was associated with its absence.

### Baseline classification results replicate original findings

As baseline, we considered the classification approach that we originally proposed in [14] and that was then used for different tasks such as detection of microbial signatures linked to colorectal cancer (CRC) from human metagenomes [39], characterization of the oral microbiome in dental implant diseases [47], and identification of changes associated with dietary interventional studies [55]. More specifically, we considered a RF classifier applied on the species-level relative abundance profiles, and evaluated classification accuracies in terms of multiple metrics (i.e., area under the ROC curve (AUC), area under the precision-recall curve (AUPRC), precision, recall, and F1) using a cross-validation (CV) approach (see Materials and Methods). We obtained variable accuracies ranging from 0.56 (in terms of AUC) for hypertension in the LiJ\_2017 dataset [44] to 0.99 for IBD in the IjazUZ\_2017 dataset [45], with an average AUC across the 25 case-control studies equal to 0.83 (S4 Table). Such values were in line with what reported in the original publications, although a fair comparison is difficult to be performed due to differences in terms of adopted algorithms and input features. On the 17 publications that reported classification results on the same samples here considered, we obtained an average AUC of 0.80 in comparison to the average of 0.83 reported in the original publications (S4 Table).

# Degradation from species-level relative abundance to presence/absence profiles does not worsen classification accuracies

We applied the same classification approach on the same set of samples to the presence/ absence profiles (Materials and Methods). In this way, we evaluated to which extent moving from relative abundance to presence/absence information could impact classification accuracies. Surprisingly, we observed negligible differences between the two experimental settings (Figs 1B, 1C and S1 and S2 Table). In both cases (i.e., using presence/absence or relative abundance profiles), we obtained an average AUC of 0.83 (AUPRC = 0.83) across the 25 casecontrol studies, with AUC and AUPRC values strongly correlated (S2 Fig; Spearman correlation = 0.918). Some variations were observed at dataset-level (relative abundance outperformed presence/absence at a maximum of 0.06 in terms of AUC in the RaymondF\_2016 dataset [34], while the opposite case was verified in YeZ\_2018 [33] for an AUC difference of 0.07), however these were likely due to random perturbations and in none of the cases they were associated with statistically significant differences (p > 0.05, <u>S2 Table</u>). This was also confirmed in terms of the other metrics of comparison (i.e., precision, recall, and F1), with no significant differences between the two profile types (S2 Table). In a similar setting, we performed body site discrimination (oral vs stool samples) in the HMP dataset [10], with a value of AUC equal to 1.00 for both profile types. Therefore, such findings suggested that it was more the presence of same taxa rather than their actual relative abundance to be relevant for discrimination purposes.

We extended this analysis to 16S rRNA samples. More specifically, we considered the same set of 30 case-control studies for a total of 4,026 samples that were originally collected and analysed in [13] (Fig 2A and S1 Table). We applied the same pre-processing procedure adopted in [13] (Materials and Methods), and performed the prediction tasks by adopting the classification pipeline already considered for shotgun data. We obtained results similar to the ones presented in [13] on the genus-level relative abundance profiles (average AUC across the 30

datasets equal to 0.76 and 0.74 in our analysis and in [13], respectively) (**S5 Table**), although some differences could occur due to the different code implementations. By degrading relative abundance to presence/absence profile, we obtained few differences in the classification results between the two profile types (**Figs 2B, 2C and S4 and S5 Table**). Average AUC across the 30 studies was quite close (0.76 for relative abundance and 0.75 for presence/absence profiles), with differences that were statistically significant in only 3 out of 30 cases (**S5 Table**). Such differences, albeit impacting a limited number of datasets, may be due to the coarser taxonomic resolution and the higher noise component associated with 16S data.

# Statistically significant taxa are consistent between relative abundance and presence/absence profiles

We extended the analysis from classification to identification of differentially abundant/present taxa (i.e., possible biomarkers) through statistical testing (Materials and Methods). By comparing the sets of statistically significant species in the different case-control studies (q < 0.05; using Mann-Whitney U test for relative abundance and Fisher exact test for presence/absence profiles, both corrected through false detection rate (FDR), <u>S6 Table</u>) we found similar numbers (Fig 1D and S7 Table), with values more driven by disease and dataset types than average number of reads (S5 Fig). On average, we found 39 and 32 significant species from relative abundance and presence/absence profiles, respectively. We may hypothesize that diseases that rely on rare biomarkers are less affected by degradation to presence/absence profiles than the ones that are characterized by stronger community shifts in abundant and prevalent taxa. Although this is not sufficiently supported by our data, further investigation in this direction is warranted.

On a per dataset basis, p-values associated with statistically significant species correlated well between relative abundance and presence/absence profiles (S6 Fig). This was reflected also by the high percentage of taxa (78%) that were detected as significant in both cases, which was further confirmed by performing hierarchical clustering on the set of statistically significant taxa coming from relative abundance and presence/absence profiles (S7 Fig). Conversely, we identified discrepancies between case-enriched and control-enriched taxa in only 1.74% of the statistically significant features, which were coming from just 5 of the 24 analysed datasets (S8 Fig). Moreover, we didn't identify any taxa for which the two tests disagreed across datasets (S8 Fig).

Focusing on the gut microbiome datasets, we also identified the species that were mostly associated with disease or health (**S7 Fig**). The species most enriched in cases was *Clostridium bolteae* (significant in 78% of the diseases), followed by *Streptococcus anginosus group* (55%), *Ruthenibacterium lactatiformans* (55%), *Hungatella hathewayi* (55%), and *Eisenbergiella tayi* (55%) with all of them already reported in the literature as possible biomarkers for different disease conditions [6,13,39,41,56]. Similarly, species most enriched in controls were *Anaerostipes hadrus* (significant in 66% of the diseases), *Roseburia faecis* (55%), *Roseburia intestinalis* (55%), *Prevotella copri* (44%), and *Eubacterium hallii* (44%) [6,10,39,57].

Consistence between relative abundance and presence/absence outcomes was finally obtained on the 16S data, with 20 and 15 genera that were found to be significant on average from relative abundance and presence/absence profiles, respectively (Fig 2D and S8 Table).

# Relative abundance values lower than 0.001% do not impact classification outcomes

We evaluated how different values in thresholding relative abundance profiles could impact classification results. We thresholded the abundances at different values (i.e., moving from a

threshold equal to 0%—which corresponded to the presence/absence scenario discussed in the previous section-to 0.0001%, 0.001%, 0.01%, and 0.1%, Materials and Methods), meaning that values below the chosen threshold were forced to zero. We did not observe changes in the classification accuracy when the threshold was set to 0.0001% and 0.001% (Figs 3A and S3A and S2 Table). In both cases, we got an average AUC = 0.83 across the 25 case-control studies as obtained on the relative abundance profiles and using a threshold equal to zero, and no statistically significant differences were found. This was reflected by the number of statistically significant species (Fig 3B and S7 Table) that decreased very marginally from 32 (average value by considering 0% or 0.0001% as threshold) to 31 (threshold = 0.001%). Although very low abundant species may be actual biomarkers, they did not contribute to improving classification accuracies which was likely due to the impossibility to estimate their presence and relative abundance properly as being below or close to the limit of detection, which we quantified in this setting to be around 0.001% (with an average number of reads across our considered metagenomes equal to 47.5M). Major differences were obtained when thresholding at higher values (i.e., 0.01% and 0.1%). In these cases, average AUC decreased to 0.81 (threshold = 0.01%) and 0.78 (threshold = 0.1%), with significant differences in 3 and 6 cases, respectively.

Results on rarefied reads (Materials and Methods) showed, as expected, a slight decrease in terms of classification accuracies and number of detected biomarkers with respect to the original data set, although patterns in function of the thresholding value when going from relative abundance to presence/absence data were confirmed (S9 Table).

#### Coarser taxonomic levels are less robust to profile degradation

We further tested to which extent classification accuracy was affected by the taxonomic resolution level considered to feed the classifier. By considering original relative abundance profiles, average AUC moved from 0.83 (species-level resolution) to 0.80 (with 3 statistically significant cases), 0.78 (6), and 0.76 (11) for genus, family, and order levels, respectively (S10 Table). Such differences, albeit not too strong, suggested species as "optimal" level to optimize classification accuracies, with further improvements that may be obtained—although not tested here due to methodological limitations—with sub-species- or strain-level resolutions.

Similarly, we compared classification accuracies between relative abundance and presence/ absence profiles at different taxonomic levels. While no differences were obtained at specieslevel (as already discussed in Fig 1), we observed that coarser resolutions brought increasing AUC differences (Figs 4 and S3B and S11 Table). An average AUC difference of 0.022, 0.041, and 0.061 was obtained for genus, family, and order, respectively (with 0, 1, and 2 statistically significant cases, respectively). Similar patterns were observed in terms of number of statistically significant features (S7 Table).

#### Findings are robust to cross-study analysis and to the classifier choice

We applied the same approach on a cross-study setting. We considered the ten independent metagenomics studies associated with CRC for a total of 1313 samples (**Table 1**) and applied a leave-one-dataset-out (LODO) approach in which the model was built on all datasets but the single dataset used for testing (**Materials and Methods**). As previously reported [39,41], we observed an overall moderate decrease of the accuracy when moving from CV (average AUC equal to 0.80) to LODO (average AUC equal to 0.76; **S9 Fig and S12 Table**). More importantly, we confirmed previous findings in terms of stability of the accuracy when moving from relative abundance to presence/absence profiles at species-level resolution (**Fig 6A**). The average AUC remained stable at 0.76 for the presence/absence profiles at threshold equal to 0%, 0.0001%, and 0.001%, while it decreased to 0.74 and 0.73 when thresholding at 0.01% and





https://doi.org/10.1371/journal.pcbi.1010066.g006

0.1%, respectively. We also confirmed that the better taxonomic resolution was associated with smaller classification performance differences between relative abundance and presence/ absence data (Fig 6B and 6C).

We finally tested if the choice of the classification method could impact the findings described in the previous sections in terms of degradation from relative abundance to presence/absence profiles. First, we confirmed [14] the superiority of RF with respect to other four classification methods (i.e., Lasso [58], Elastic Net [59], and support vector machines (SVMs) with linear and RBF kernels [60]) on both relative abundance (S10A Fig and S13 Table) and presence/absence profiles (S10B Fig and S13 Table), and this was also verified on 16S rRNA data (S3 Table). On average, thresholding of relative abundance values did not negatively impact classification accuracies, instead it generally improved results in a quite unexpected way (Figs 5 and S3C). Higher differences were observed for Lasso, with an average AUC equal to 0.79 and 0.72 for presence/absence and relative abundance data, respectively, and the same pattern was obtained for the other classifier methods (with an average difference in terms of AUC equal to 0.05, 0.03, and 0.02 for ENet, LSVM, and SVM, respectively). We observed a greater variability of the classification accuracies with respect to what was observed for RF classification. In fact, we obtained statistically significant differences in Lasso, ENet, LSVM, and SVM studies for 10, 6, 5, and 6, respectively, however always in majority in favour of the presence/absence data. We therefore conclude that, despite a few differences occurred in a limited number of cases, maximization of classification accuracies was generally made possible through presence/absence profiles.

# Conclusions

In the present study, we conducted a meta-analysis on 25 publicly available datasets spanning more than 4,000 shotgun metagenomes and associated with different case-control studies. By applying species-level taxonomic profiling and machine-learning based classification approaches based on state-of-the-art methodologies we demonstrated that the presence of microbial taxa is sufficient to maximize classification accuracies. This was accomplished by degrading original relative abundance data to presence/absence profiles by considering different threshold values. We estimated a value of 0.001% in terms of relative abundance as limit of detection, meaning that although very low abundant species may be actual biomarkers they were not useful to improve classification accuracy. Results were robust to the choice of the classifier. This was obtained by considering different traditional classification algorithms that are designed for continuous data and potentially "suboptimal" when applied on binary data. This actually reinforces our findings, meaning that accuracies may be even better when models on presence/absence profiles are trained using classifiers more designed for binary data. Moreover, although doing an extensive evaluation of existing classifiers is out of the scope of the present study, maximization of classification accuracies may be reached by adopting other classification approaches including the ones specifically proposed for microbiome data analysis [61,62]. Findings were finally extended from cross validation to cross study analysis and confirmed on 16S rRNA data associated with a compendium of more than 4,000 samples coming from 30 public studies.

The growing literature aiming at identifying microbial biomarkers for different diseases opened the possibility to build non-invasive diagnostic tools from microbiome data. To this purpose, much superior accuracy can be achieved by considering multi-feature rather than single biomarkers diagnostic models, and in which machine learning-based classification approaches have a fundamental role in building such models. Moreover, maximal accuracy can usually be achieved by using a limited number of features (in the order of ten or twenty). Such findings recently presented in the literature in addition to outcomes of our study, which suggest that the detection of microbial taxa is sufficient to maximize classification accuracies, are important steps toward the development of fast and inexpensive tests applied on stool samples for diagnostic purposes.

# **Supporting information**

S1 Table. Summary of the 30 classification tasks derived from 16S rRNA datasets for casecontrol prediction. ASD: Autism spectrum disorder, CD: Crohn disease, CDI: Clostridium difficile infection, CIRR: Cirrhosis, MHE: Minimal hepatic encephalopathy, CRC: Colorectal cancer, EDD: enteric diarrheal disease, HIV: human immunodeficiency virus, NASH: non-alcoholic steatohepatitis, OB: obesity, PAR: Parkinson's disease, PSA: psoriatic arthritis, RA: Rheumatoid arthritis, T1D: type-1 diabetes, UC: ulcerative colitis. Non-CDI controls are patients with diarrhea who tested negative for *C. difficile* infection. (XLSX)

**S2 Table. Results obtained from the classification process done on the shotgun datasets.** Comparison in terms of AUC, AUPRC, F1, precision, recall between relative abundance and presence/absence profiles at different threshold levels. Results are obtained using RF classification at the species-level taxonomic resolution. (XLSX)

S3 Table. Comparison in terms of AUC between relative abundance and presence/absence profiles with different classification algorithms (RF: Random Forest; Lasso; ENet: Elastic Net; LSVM: SVM with linear kernel; SVM: SVM with RBF kernel). (XLSX)

S4 Table. Comparison in terms of AUC between our results (using RF classification on the relative abundance profiles) and the ones reported in the original publications. In most of the cases, different classifier algorithms and/or input features were used in the original analysis. Original papers that did not conduct a classification analysis are not included in this table. (XLSX)

**S5 Table. Results obtained from the classification process done on the 16s datasets.** Comparison in terms of AUC, AUPRC, F1, precision, recall between relative abundance and presence/absence profiles at different threshold levels. Results are obtained using RF classification at the species-level taxonomic resolution. (XLSX)

S6 Table. P-values (after FDR correction) obtained by testing differences in abundance of each species between controls and cases.

## (XLSX)

S7 Table. Number of statistically significant taxa ( $q \le 0.05$ ) between cases and controls for each shotgun dataset and at varying input features (relative abundance vs presence/ absence profiles) and taxonomic level. (XLSX)

S8 Table. Number of statistically significant taxa (q < = 0.05) between cases and controls for each 16s dataset and at varying input features (relative abundance vs presence/absence profiles).

(XLSX)

S9 Table. Results obtained on three selected shotgun datasets after rarefying metagenomes at 1M reads. Comparison in terms of AUC, F1, precision, recall, in addition to number of statistically significant taxa (q < = 0.05), between the results obtained classifying on the abundances matrix and the classification made on the presence/absence boolean matrix at different taxonomic levels (only at species level). (XLSX)

**S10 Table. Results obtained from the classification process done on the shotgun datasets.** Comparison in terms of AUC between the results obtained classifying at different taxonomic resolution levels. The results are obtained using the RF classifier on the relative abundances matrixes.

(XLSX)

**S11 Table. Results obtained from the classification process done on the shotgun dataset.** Comparison in terms of AUC, F1, precision, recall between the results obtained classifying on the abundances matrix and the classification made on the presence/absence boolean matrix at different taxonomic levels (species, genus, etc). (XLSX)

**S12 Table. Results obtained by the LODO classification for datasets associated with CRC.** Comparison in terms of AUC obtained classifying thresholding the dataset at different levels and at different taxonomic levels. (XLSX)

S13 Table. Comparison in terms of AUC, F1, precision, recall between the results obtained from different classifiers on the relative abundances matrix and on the presence absence boolean matrix (only at species level). (XLSX)

S1 Fig. Classification accuracies are robust to degradation from species-level relative abundance to presence/absence profiles in shotgun datasets. Comparison in terms of AUC between presence/absence and relative abundance profiles for the 25 case-control shotgun datasets.

(PNG)

**S2 Fig. AUC correlates well with AUPRC.** Comparison in terms of classification accuracies between AUC (area under the curve) and AUPRC (area under the precision-recall curve) for the 25 case-control shotgun datasets and by considering relative abundance (in blue; Spearman correlation = 0.889) and presence/absence (in red; Spearman correlation = 0.918) profiles. (PNG)

S3 Fig. Classification accuracies are robust to degradation from species-level relative abundance to presence/absence profiles in shotgun datasets. Comparison in terms of AUC between presence/absence and relative abundance profiles for the 25 case-control shotgun datasets by (A) thresholding at different relative abundance values (ranging from 0% to 0.1%), (B) changing taxonomic resolution (from species to order level), and (C) changing classification algorithm.

#### (PNG)

**S4 Fig. Classification accuracies are robust to degradation from species-level relative abundance to presence/absence profiles in 16S rRNA datasets.** Comparison in terms of AUC between presence/absence and relative abundance profiles for the 30 case-control 16 rRNA datasets. (PNG)

**S5 Fig. Number of differentially abundant species has weak correlation with the average number of reads.** Each dot represents one of the 26 case-control shotgun studies. The number of statistically significant species is computed on relative abundance profiles. (PNG)

S6 Fig. P-values associated with statistically significant species correlate well between relative abundance and presence/absence profiles. Each dot represents a different taxa (i.e., species) and we report only species significant in at least one of the two data types. Only datasets with at least ten data points are shown.

(PNG)

S7 Fig. Statistically significant taxa are consistent between relative abundance and presence/absence data on a per dataset basis. Heatmap generated on the p-values (after FDR correction; p > 0.05 in grey) obtained by applying statistical tests on the case-control metagenomic datasets. Only the 18 datasets with at least one discriminative taxa are reported. Left-most colorbar identifies the taxonomic class of each taxa. The two right-most colorbars indicate the percentage of diseases for which the species resulted to be enriched in controls (in green) and in cases (in red). This percentage is computed on a per disease basis, when multiple datasets are available for the same disease, the taxa is considered significant when detected as significant in at least one dataset.

(PNG)

**S8 Fig. Statistically significant taxa from relative abundance and presence/absence profiles did not disagree across datasets.** We identified discrepancies between case-enriched and control-enriched taxa derived from relative abundance and presence/absence data in only 1.74% of the statistically significant features, which were coming from just 5 datasets. No taxa disagreed across datasets. (PNG)

**S9 Fig. Degradation of relative abundance profiles has a limited impact on both CV and LODO classification.** AUC scores using RF as back-end classifiers on species-level relative abundance and corresponding presence/absence profiles in CV and LODO settings. (PNG)

**S10 Fig. RFs generally outperform other classifiers.** Results on the 25 case-control shotgun studies by considering different classification algorithms. Difference in AUC between RFs and other classification methods on (**A**) the relative abundance and (**B**) the presence/absence profiles. A positive value indicates that the comparison method outperforms RFs. (PNG)

# Author Contributions

Conceptualization: Edoardo Pasolli. Data curation: Renato Giliberti. Formal analysis: Renato Giliberti, Edoardo Pasolli. Funding acquisition: Danilo Ercolini, Edoardo Pasolli. Investigation: Renato Giliberti. Methodology: Renato Giliberti, Edoardo Pasolli. Project administration: Edoardo Pasolli. Software: Renato Giliberti, Edoardo Pasolli. Supervision: Edoardo Pasolli. Validation: Renato Giliberti, Sara Cavaliere, Italia Elisa Mauriello. Visualization: Renato Giliberti. Writing - review & editing: Renato Giliberti, Danilo Ercolini, Edoardo Pasolli.

### References

- 1. Lynch SV, Pedersen O. The Human Intestinal Microbiome in Health and Disease. N Engl J Med. 2016; 375: 2369–2379. https://doi.org/10.1056/NEJMra1600266 PMID; 27974040
- Zhou Y-H, Gallins P. A Review and Tutorial of Machine Learning Methods for Microbiome Host Trait Prediction. Front Genet. 2019; 10: 579. https://doi.org/10.3389/fgene.2019.00579 PMID: 31293616
- 3. Cui H, Zhang X. Alignment-free supervised classification of metagenomes by recursive SVM. BMC Genomics. 2013; 14: 641. https://doi.org/10.1186/1471-2164-14-641 PMID: 24053649
- 4. Sze MA, Schloss PD. Looking for a Signal in the Noise: Revisiting Obesity and the Microbiome. MBio. 2016; 7. https://doi.org/10.1128/mBio.01018-16 PMID: 27555308
- Vatanen T, Franzosa EA, Schwager R, Tripathi S, Arthur TD, Vehik K, et al. The human gut microbiome in early-onset type 1 diabetes from the TEDDY study. Nature. 2018; 562: 589–594. https://doi.org/10. 1038/s41586-018-0620-2 PMID: 30356183
- Zeller G, Tap J, Voigt AY, Sunagawa S, Kultima JR, Costea PI, et al. Potential of fecal microbiota for early-stage detection of colorectal cancer. Mol Syst Biol. 2014; 10: 766. https://doi.org/10.15252/msb. 20145645 PMID: 25432777
- Eloe-Fadrosh EA, Rasko DA. The human microbiome: from symbiosis to pathogenesis. Annu Rev Med. 2013; 64: 145–163. https://doi.org/10.1146/annurev-med-010312-133513 PMID: 23327521
- McCoubrey LE, Elbadawi M, Orlu M, Gaisford S, Basit AW. Harnessing machine learning for development of microbiome therapeutics. Gut Microbes. 2021; 13: 1–20. https://doi.org/10.1080/19490976. 2021.1872323 PMID: 33522391
- Qin J, Li R, Raes J, Arumugam M, Burgdorf KS, Manichanh C, et al. A human gut microbial gene catalogue established by metagenomic sequencing. Nature. 2010; 464: 59–65. https://doi.org/10.1038/ nature08821 PMID: 20203603
- Human Microbiome Project Consortium. Structure, function and diversity of the healthy human microbiome. Nature. 2012; 486: 207–214. https://doi.org/10.1038/nature11234 PMID: 22699609
- 11. Lozupone CA, Stombaugh J, Gonzalez A, Ackermann G, Wendel D, Vázquez-Baeza Y, et al. Metaanalyses of studies of the human microbiota. Genome Res. 2013; 23: 1704–1714. https://doi.org/10. 1101/gr.151803.112 PMID: 23861384
- 12. Statnikov A, Henaff M, Narendra V, Konganti K, Li Z, Yang L, et al. A comprehensive evaluation of multicategory classification methods for microbiomic data. Microbiome. 2013; 1: 11. <u>https://doi.org/10.1186/</u> 2049-2618-1-11 PMID: 24456583
- Duvallet C, Gibbons SM, Gurry T, Irizarry RA, Alm EJ. Meta-analysis of gut microbiome studies identifies disease-specific and shared responses. Nat Commun. 2017; 8: 1784. https://doi.org/10.1038/ s41467-017-01973-8 PMID: 29209090
- Pasolli E, Truong DT, Malik F, Waldron L, Segata N. Machine Learning Meta-analysis of Large Metagenomic Datasets: Tools and Biological Insights. PLoS Comput Biol. 2016; 12: e1004977. <u>https://doi.org/</u> 10.1371/journal.pcbi.1004977 PMID: 27400279
- Armour CR, Nayfach S, Pollard KS, Sharpton TJ. A Metagenomic Meta-analysis Reveals Functional Signatures of Health and Disease in the Human Gut Microbiome. mSystems. 2019; 4. https://doi.org/ 10.1128/mSystems.00332-18 PMID: 31098399
- Vangay P, Hillmann BM, Knights D. Microbiome Learning Repo (ML Repo): A public repository of microbiome regression and classification tasks. Gigascience. 2019; 8. <u>https://doi.org/10.1093/gigascience/ giz042 PMID: 31042284</u>
- Wirbel J, Zych K, Essex M, Karcher N, Kartal E, Salazar G, et al. Microbiome meta-analysis and crossdisease comparison enabled by the SIAMCAT machine learning toolbox. Genome Biol. 2021; 22: 93. https://doi.org/10.1186/s13059-021-02306-1 PMID: 33785070
- Moreno-Indias I, Lahti L, Nedyalkova M, Elbere I, Roshchupkin G, Adilovic M, et al. Statistical and Machine Learning Techniques in Human Microbiome Studies: Contemporary Challenges and Solutions. Front Microbiol. 2021; 12: 635781. https://doi.org/10.3389/fmicb.2021.635781 PMID: 33692771
- Marcos-Zambrano LJ, Karaduzovic-Hadziabdic K, Loncar Turukalo T, Przymus P, Trajkovik V, Aasmets O, et al. Applications of Machine Learning in Human Microbiome Studies: A Review on Feature Selection, Biomarker Identification, Disease Prediction and Treatment. Front Microbiol. 2021; 12: 634511. https://doi.org/10.3389/fmicb.2021.634511 PMID: 33737920

- Ditzler G, Morrison JC, Lan Y, Rosen GL. Fizzy: feature subset selection for metagenomics. BMC Bioinformatics. 2015; 16: 358. https://doi.org/10.1186/s12859-015-0793-8 PMID: 26538306
- Oudah M, Henschel A. Taxonomy-aware feature engineering for microbiome classification. BMC Bioinformatics. 2018; 19: 227. https://doi.org/10.1186/s12859-018-2205-3 PMID: 29907097
- Wu H, Cai L, Li D, Wang X, Zhao S, Zou F, et al. Metagenomics Biomarkers Selected for Prediction of Three Different Diseases in Chinese Population. Biomed Res Int. 2018; 2018; 2936257. https://doi.org/ 10.1155/2018/2936257 PMID: 29568746
- Bang S, Yoo D, Kim S-J, Jhang S, Cho S, Kim H. Establishment and evaluation of prediction model for multiple disease classification based on gut microbial data. Sci Rep. 2019; 9: 10189. https://doi.org/10. 1038/s41598-019-46249-x PMID: 31308384
- 24. Wang X-W, Liu Y-Y. Comparative study of classifiers for human microbiome data. Medicine in Microecology. 2020; 4: 100013. https://doi.org/10.1016/j.medmic.2020.100013 PMID: 34368751
- LaPierre N, Ju CJ-T, Zhou G, Wang W. MetaPheno: A critical evaluation of deep learning and machine learning in metagenome-based disease prediction. Methods. 2019; 166: 74–82. https://doi.org/10. 1016/j.ymeth.2019.03.003 PMID: 30885720
- López CD, Vidaki A, Ralf A, González DM, Radjabzadeh D, Kraaij R, et al. Novel taxonomy-independent deep learning microbiome approach allows for accurate classification of different forensically relevant human epithelial materials. Forensic Science International: Genetics. 2019. pp. 72–82. <a href="https://doi.org/10.1016/j.fsigen.2019.03.015">https://doi.org/10.1016/j.fsigen.2019.03.015</a> PMID: 31003081
- 27. Oh M, Zhang L. DeepMicro: deep representation learning for disease prediction based on microbiome data. Sci Rep. 2020; 10: 6026. https://doi.org/10.1038/s41598-020-63159-5 PMID: 32265477
- Ching T, Himmelstein DS, Beaulieu-Jones BK, Kalinin AA, Do BT, Way GP, et al. Opportunities and obstacles for deep learning in biology and medicine. J R Soc Interface. 2018; 15. https://doi.org/10. 1098/rsif.2017.0387 PMID: 29618526
- Beghini F, McIver LJ, Blanco-Míguez A, Dubois L, Asnicar F, Maharjan S, et al. Integrating taxonomic, functional, and strain-level profiling of diverse microbial communities with bioBakery 3. Elife. 2021; 10. https://doi.org/10.7554/eLife.65088 PMID: 33944776
- Pasolli E, Schiffer L, Manghi P, Renson A, Obenchain V, Truong DT, et al. Accessible, curated metagenomic data through ExperimentHub. Nat Methods. 2017; 14: 1023–1024. https://doi.org/10.1038/ nmeth.4468 PMID: 29088129
- Jie Z, Xia H, Zhong S-L, Feng Q, Li S, Liang S, et al. The gut microbiome in atherosclerotic cardiovascular disease. Nat Commun. 2017; 8: 845. https://doi.org/10.1038/s41467-017-00900-1 PMID: 29018189
- Chng KR, Tay ASL, Li C, Ng AHQ, Wang J, Suri BK, et al. Whole metagenome profiling reveals skin microbiome-dependent susceptibility to atopic dermatitis flare. Nat Microbiol. 2016; 1: 16106. <u>https://</u> doi.org/10.1038/nmicrobiol.2016.106 PMID: 27562258
- **33.** Ye Z, Zhang N, Wu C, Zhang X, Wang Q, Huang X, et al. A metagenomic study of the gut microbiome in Behcet's disease. Microbiome. 2018; 6: 135. <u>https://doi.org/10.1186/s40168-018-0520-6</u> PMID: 30077182
- Raymond F, Ouameur AA, Déraspe M, Iqbal N, Gingras H, Dridi B, et al. The initial state of the human gut microbiome determines its reshaping by antibiotics. ISME J. 2016; 10: 707–720. <u>https://doi.org/10. 1038/ismej.2015.148 PMID: 26359913</u>
- 35. Qin N, Yang F, Li A, Prifti E, Chen Y, Shao L, et al. Alterations of the human gut microbiome in liver cirrhosis. Nature. 2014; 513: 59–64. https://doi.org/10.1038/nature13568 PMID: 25079328
- Feng Q, Liang S, Jia H, Stadlmayr A, Tang L, Lan Z, et al. Gut microbiome development along the colorectal adenoma-carcinoma sequence. Nat Commun. 2015; 6: 6528. <u>https://doi.org/10.1038/</u> ncomms7528 PMID: 25758642
- Gupta A, Dhakan DB, Maji A, Saxena R, P K VP, Mahajan S, et al. Association of Flavonifractor plautii, a Flavonoid-Degrading Bacterium, with the Gut Microbiome of Colorectal Cancer Patients in India. mSystems. 2019; 4. https://doi.org/10.1128/mSystems.00438-19 PMID: 31719139
- Hannigan GD, Duhaime MB, Ruffin MT 4th, Koumpouras CC, Schloss PD. Diagnostic Potential and Interactive Dynamics of the Colorectal Cancer Virome. MBio. 2018; 9. https://doi.org/10.1128/mBio. 02248-18 PMID: 30459201
- Thomas AM, Manghi P, Asnicar F, Pasolli E, Armanini F, Zolfo M, et al. Metagenomic analysis of colorectal cancer datasets identifies cross-cohort microbial diagnostic signatures and a link with choline degradation. Nat Med. 2019; 25: 667–678. https://doi.org/10.1038/s41591-019-0405-7 PMID: 30936548
- 40. Vogtmann E, Hua X, Zeller G, Sunagawa S, Voigt AY, Hercog R, et al. Colorectal Cancer and the Human Gut Microbiome: Reproducibility with Whole-Genome Shotgun Sequencing. PLoS One. 2016; 11: e0155362. https://doi.org/10.1371/journal.pone.0155362 PMID: 27171425

- Wirbel J, Pyl PT, Kartal E, Zych K, Kashani A, Milanese A, et al. Meta-analysis of fecal metagenomes reveals global microbial signatures that are specific for colorectal cancer. Nat Med. 2019; 25: 679–689. https://doi.org/10.1038/s41591-019-0406-6 PMID: 30936547
- Yachida S, Mizutani S, Shiroma H, Shiba S, Nakajima T, Sakamoto T, et al. Metagenomic and metabolomic analyses reveal distinct stage-specific phenotypes of the gut microbiota in colorectal cancer. Nat Med. 2019; 25: 968–976. https://doi.org/10.1038/s41591-019-0458-7 PMID; 31171880
- Yu J, Feng Q, Wong SH, Zhang D, Liang QY, Qin Y, et al. Metagenomic analysis of faecal microbiome as a tool towards targeted non-invasive biomarkers for colorectal cancer. Gut. 2017; 66: 70–78. https:// doi.org/10.1136/gutjnl-2015-309800 PMID: 26408641
- Li J, Zhao F, Wang Y, Chen J, Tao J, Tian G, et al. Gut microbiota dysbiosis contributes to the development of hypertension. Microbiome. 2017; 5: 14. https://doi.org/10.1186/s40168-016-0222-x PMID: 28143587
- 45. Ijaz UZ, Quince C, Hanske L, Loman N, Calus ST, Bertz M, et al. The distinct features of microbial "dysbiosis" of Crohn's disease do not occur to the same extent in their unaffected, genetically-linked kindred. PLoS One. 2017; 12: e0172605. https://doi.org/10.1371/journal.pone.0172605 PMID: 28222161
- 46. Nielsen HB, Almeida M, Juncker AS, Rasmussen S, Li J, Sunagawa S, et al. Identification and assembly of genomes and genetic elements in complex metagenomic samples without using reference genomes. Nat Biotechnol. 2014; 32: 822–828. https://doi.org/10.1038/nbt.2939 PMID: 24997787
- Ghensi P, Manghi P, Zolfo M, Armanini F, Pasolli E, Bolzan M, et al. Strong oral plaque microbiome signatures for dental implant diseases identified by strain-resolution metagenomics. NPJ Biofilms Microbiomes. 2020; 6: 47. https://doi.org/10.1038/s41522-020-00155-7 PMID: 33127901
- Castro-Nallar E, Bendall ML, Pérez-Losada M, Sabuncyan S, Severance EG, Dickerson FB, et al. Composition, taxonomy and functional diversity of the oropharynx microbiome in individuals with schizophrenia and controls. PeerJ. 2015; 3: e1140. https://doi.org/10.7717/peerj.1140 PMID: 26336637
- Heintz-Buschart A, May P, Laczny CC, Lebrun LA, Bellora C, Krishna A, et al. Integrated multi-omics of the human gut microbiome in a case study of familial type 1 diabetes. Nat Microbiol. 2016; 2: 16180. https://doi.org/10.1038/nmicrobiol.2016.180 PMID: 27723761
- Kostic AD, Gevers D, Siljander H, Vatanen T, Hyötyläinen T, Hämäläinen A-M, et al. The dynamics of the human infant gut microbiome in development and in progression toward type 1 diabetes. Cell Host Microbe. 2015; 17: 260–273. https://doi.org/10.1016/j.chom.2015.01.001 PMID: 25662751
- Karlsson FH, Tremaroli V, Nookaew I, Bergström G, Behre CJ, Fagerberg B, et al. Gut metagenome in European women with normal, impaired and diabetic glucose control. Nature. 2013; 498: 99–103. https://doi.org/10.1038/nature12198 PMID: 23719380
- Qin J, Li Y, Cai Z, Li S, Zhu J, Zhang F, et al. A metagenome-wide association study of gut microbiota in type 2 diabetes. Nature. 2012; 490: 55–60. https://doi.org/10.1038/nature11450 PMID: 23023125
- Morton JT, Marotz C, Washburne A, Silverman J, Zaramela LS, Edlund A, et al. Establishing microbial composition measurement standards with reference frames. Nat Commun. 2019; 10: 2719. https://doi. org/10.1038/s41467-019-10656-5 PMID: 31222023
- Ling W, Zhao N, Plantinga AM, Launer LJ, Fodor AA, Meyer KA, et al. Powerful and robust nonparametric association testing for microbiome data via a zero-inflated quantile approach (ZINQ). Microbiome. 2021; 9: 181. https://doi.org/10.1186/s40168-021-01129-3 PMID: 34474689
- Meslier V, Laiola M, Roager HM, De Filippis F, Roume H, Quinquis B, et al. Mediterranean diet intervention in overweight and obese subjects lowers plasma cholesterol and causes changes in the gut microbiome and metabolome independently of energy intake. Gut. 2020; 69: 1258–1268. <u>https://doi.org/10.1136/gutjnl-2019-320438 PMID</u>: 32075887
- Pandit L, Cox LM, Malli C, D'Cunha A, Rooney T, Lokhande H, et al. is elevated in neuromyelitis optica spectrum disorder in India and shares sequence similarity with AQP4. Neurol Neuroimmunol Neuroinflamm. 2021; 8. https://doi.org/10.1212/NXI.000000000000907 PMID: 33148687
- Tamanai-Shacoori Z, Smida I, Bousarghin L, Loreal O, Meuric V, Fong SB, et al. Roseburia spp.: a marker of health? Future Microbiol. 2017; 12: 157–170. <u>https://doi.org/10.2217/fmb-2016-0130</u> PMID: 28139139
- 58. Tibshirani R. Regression shrinkage and selection via the lasso. J R Stat Soc. 1996; 58: 267–288.
- Zou H, Hastie T. Regularization and variable selection via the elastic net. J R Stat Soc Series B Stat Methodol. 2005; 67: 301–320.
- 60. Cortes C, Vapnik V. Support-Vector Networks. Mach Learn. 1995; 20: 273-297.
- Reiman D, Metwally AA, Sun J, Dai Y. PopPhy-CNN: A Phylogenetic Tree Embedded Architecture for Convolutional Neural Networks to Predict Host Phenotype From Metagenomic Data. IEEE J Biomed Health Inform. 2020; 24: 2993–3001. https://doi.org/10.1109/JBHI.2020.2993761 PMID: 32396115

**62.** Rahman MA, Rangwala H. IDMIL: an alignment-free Interpretable Deep Multiple Instance Learning (MIL) for predicting disease from whole-metagenomic data. Bioinformatics. 2020; 36: i39–i47. https://doi.org/10.1093/bioinformatics/btaa477 PMID: 32657370